Author's response to reviews

Title: An automated in vitro model for the evaluation of ultrasound modalities measuring myocardial deformation

Authors:

Albin Stigö (albin.stigo@gmail.com)
Peter Johansen (peter.johansen@ki.au.dk)
Morten Ø Jensen (morten.jensen@ki.au.dk)
Kim Sivesgaard (kim.sivesgaard@gmail.com)
Hans Nygaard (nygaard@ki.au.dk)
Erik Sloth (sloth@dadlnet.dk)

Version: 4 Date: 26 August 2010

Author's response to reviews:

Major revisions:
Please present all text in black:
- All text color changed to black.

Authors contributions: Please format these as a paragraph rather than a list.
- It has been reformatted.